Author’s response to reviews

Title: A dynamic model and some strategies on how to prevent and control hepatitis C in mainland China

Authors:

Wanru Jia (1575616653@qq.com)
Jie Weng (1562177414@qq.com)
Cong Fang (1213139945@qq.com)
Yong Li (gxbyl@163.com)

Version: 1 Date: 22 Jul 2018

Author’s response to reviews:

Responses to the Reviewers' Report (INFD-D-18-01089)

Dear Editors and Reviewers,

We would like to thank the editors for giving us a chance to resubmit our manuscript, and also thank the editors for giving us constructive suggestions. We will submit a new version, which has been modified according to the editor’s suggestions.

Based on the comments of editors, we give the following responses.

Declarations
Ethics approval and consent to participate
Not applicable

Consent for publication
Not applicable

Availability of data and material
The data that support the findings of this study are available from the China Information System for Disease Control and Prevention, these network direct data are completely open.
Competing interests
We declare that we have no any stake with other organizations or individuals in this article. No authors have potential conflicts of interest with reference to this work.

Funding
The work was partially supported by the National Natural Science Foundation of China (11547006, 11471133), Scientific Research Project of Hubei Provincial Department of Education (B2017039), Undergraduate Training Program of Yangtze University for Innovation and Entrepreneurship (20150094).

Authors’ Contributions
These authors contributed equally to this work. LY, JW and WJ conceptualized and designed the study, drafted the initial manuscript, and approved the final manuscript as submitted. FC and WJ analyzed the data and simulated parameters. JW and LY carried out the initial analyses, reviewed and revised the manuscript.

Acknowledgments
We would like to thank anonymous reviewers for very helpful suggestions which improved greatly this manuscript.